Author's response to reviews

Title: Human breast cancer and lymph node metastases express Gb3 and can be targeted by STxB-vectorized chemotherapeutic compounds

Authors:

Lev Stimmer (lev.stimmer@cea.fr)
Sabrina Dehay (sabdehay@yahoo.fr)
Fariba Nemati (fariba.nemati@curie.fr)
Gerald Massonnet (Gerald.Massonnet@curie.fr)
Sophie Richon (sophie.richon@parisdescartes.fr)
Didier Decaudin (didier.decaudin@curie.net)
Jerzy Klijanienko (jerzy.klijanienko@curie.net)
Ludger Johannes (ludger.johannes@curie.fr)

Version: 2
Date: 1 September 2014

Author's response to reviews:

1. As you requested, we included a continuous line and page numbering in the revised manuscript.

2. Additionally, we improved the title page. It contains now the names, institutions, countries and email addresses of all authors, and the full postal address of the corresponding author.